Author’s response to reviews

Title: A randomised controlled trial of performance review and facilitated feedback to increase implementation of healthy eating and physical activity-promoting policies and practices in centre-based childcare

Authors: 
Meghan Finch (meghan.finch@hnehealth.nsw.gov.au)
Fiona Stacey (fiona.stacey@hnehealth.nsw.gov.au)
Jannah Jones (jannah.jones@hnehealth.nsw.gov.au)
Sze Lin Yoong (serene.yoong@hnehealth.nsw.gov.au)
Alice Grady (alice.grady@hnehealth.nsw.gov.au)
Luke Wolfenden (luke.wolfenden@hnehealth.nsw.gov.au)

Version: 2 Date: 24 Jan 2019

Author’s response to reviews:

Dear Professor Straus

RE: submission to Implementation Science - IMPS-D-18-00413

“A randomised controlled trial of performance review and facilitated feedback to increase implementation of healthy eating and physical activity-promoting policies and practices in centre-based childcare”

Thank you for your correspondence relating to the above manuscript submitted to the Implementation Science. We hope that we have adequately addressed the comments raised by the handling editor regarding reducing the word limit, in particular shortening the discussion.

All new changes in the manuscript are highlighted in track changes by underlined text in the revised manuscript.